\begin{thebibliography}{10}

\bibitem{04Aitken}
CGG Aitken and D~Lucy.
\newblock {Evaluation of trace evidence in the form of multivariate data}.
\newblock {\em {JOURNAL OF THE ROYAL STATISTICAL SOCIETY SERIES C-APPLIED
  STATISTICS}}, {53}({Part 1}):{109--122}, {2004}.

\bibitem{06Aitken}
CGG Aitken, D~Lucy, G~Zadora, and JM~Curran.
\newblock {Evaluation of transfer evidence for three-level multivariate data
  with the use of graphical models}.
\newblock {\em {COMPUTATIONAL STATISTICS \& DATA ANALYSIS}},
  {50}({10}):{2571--2588}, {JUN 20} {2006}.

\bibitem{Bezdek2000}
JC~Bezdek and LI~Kuncheva.
\newblock {Some notes on twenty one (21) nearest prototype classifiers}.
\newblock In {Ferri, FJ and Inesta, JM and Amin, A and Pudil, P}, editor, {\em
  {ADVANCES IN PATTERN RECOGNITION}}, volume {1876} of {\em {LECTURE NOTES IN
  COMPUTER SCIENCE}}, pages {1--16}. {2000}.

\bibitem{Bezdek2001}
JC~Bezdek and LI~Kuncheva.
\newblock {Nearest prototype classifier designs: An experimental study}.
\newblock {\em {INTERNATIONAL JOURNAL OF INTELLIGENT SYSTEMS}},
  {16}({12}):{1445--1473}, {DEC} {2001}.

\bibitem{pca}
C.M Bishop.
\newblock {\em Pattern Recognition and Machine Learning}.
\newblock New York, NY, Springer, 2006.

\bibitem{Cano2005}
José~Ramón Cano, Francisco Herrera, and Manuel Lozano.
\newblock Stratification for scaling up evolutionary prototype selection.
\newblock {\em Pattern Recognition Letters}, 26(7):953 -- 963, 2005.

\bibitem{Chang74}
Chin-Liang Chang.
\newblock Finding prototypes for nearest neighbor classifiers.
\newblock {\em Computers, IEEE Transactions on}, C-23(11):1179--1184, Nov.
  1974.

\bibitem{Hart67}
T.~Cover and P.~Hart.
\newblock Nearest neighbor pattern classification.
\newblock {\em Information Theory, IEEE Transactions on}, 13(1):21--27, Jan
  1967.

\bibitem{Dasarathy94}
B.V. Dasarathy.
\newblock Minimal consistent set (mcs) identification for optimal nearest
  neighbor decision systems design.
\newblock {\em Systems, Man and Cybernetics, IEEE Transactions on},
  24(3):511--517, Mar 1994.

\bibitem{Devi2002}
V.~Susheela Devi and M.~Narasimha Murty.
\newblock An incremental prototype set building technique.
\newblock {\em Pattern Recognition}, 35(2):505 -- 513, 2002.

\bibitem{knn}
Richard~O. Duda and Peter E.Hart.
\newblock {\em Pattern classification and scene analysis}.
\newblock A Wiley-Interscience Publication, New York: Wiley, 1973.

\bibitem{Evett84}
Ian Evett.
\newblock A quantitative theory for interpreting transfer evidence in criminal
  cases.
\newblock {\em Applied Statistics}, 33(1):25--32, 1984.

\bibitem{Evett90}
Ian Evett and John Buckleton.
\newblock The interpretation of glass evidence. a practical approach.
\newblock {\em Journal of the Forensic Science Society}, 30(4):215--223, 1990.

\bibitem{Evett94}
Ian Evett and J.~Lambert.
\newblock Further observations on glass evidence interpretation.
\newblock {\em Science and Justice}, 35(4):283--289, 1995.

\bibitem{fastmap}
Christos Faloutsos and King-Ip Lin.
\newblock Fastmap: a fast algorithm for indexing, data-mining and visualization
  of traditional and multimedia datasets.
\newblock In {\em SIGMOD '95: Proceedings of the 1995 ACM SIGMOD international
  conference on Management of data}, pages 163--174, New York, NY, USA, 1995.
  ACM.

\bibitem{Frank+Asuncion:2010}
A.~Frank and A.~Asuncion.
\newblock {UCI} machine learning repository, 2010.

\bibitem{Garain2008}
Utpal Garain.
\newblock Prototype reduction using an artificial immune model.
\newblock {\em Pattern Anal. Appl.}, 11(3-4):353--363, 2008.

\bibitem{García2008}
Salvador García, José~Ramón Cano, and Francisco Herrera.
\newblock A memetic algorithm for evolutionary prototype selection: A scaling
  up approach.
\newblock {\em Pattern Recognition}, 41(8):2693 -- 2709, 2008.

\bibitem{Grove80}
D.M. Grove.
\newblock Interpretation of forensic evidence using a likelihood ratio.
\newblock {\em Biometrika}, 67(1):243--246, April 1980.

\bibitem{Hart68a}
P.~Hart.
\newblock The condensed nearest neighbor rule (corresp.).
\newblock {\em Information Theory, IEEE Transactions on}, 14(3):515 -- 516, may
  1968.

\bibitem{jalali08}
O.~Jalali, T.~Menzies, and M.~Feather.
\newblock Optimizing requirements decisions with keys.
\newblock In {\em Proceedings of the PROMISE 2008 Workshop (ICSE)}, 2008.
\newblock Available from \url{http://menzies.us/pdf/08keys.pdf}.

\bibitem{joll02}
I.~Jolliffe.
\newblock {\em Principal component analysis}.
\newblock Springer-Verlag, 175 Fifth Avenue, NY, USA, 2002.

\bibitem{Karslake09}
N.~Karslake, S.~Lewis, and W.~Bronswijk.
\newblock Characterisation of automotive paint clear coats by atr-fr-ir with
  subsequent chemometric analysis.
\newblock 2009.

\bibitem{Kim2003}
SW~Kim and BJ~Oommen.
\newblock {A brief taxonomy and ranking of creative prototype reduction
  schemes}.
\newblock {\em {PATTERN ANALYSIS AND APPLICATIONS}}, {6}({3}):{232--244}, {DEC}
  {2003}.

\bibitem{Kohonen90}
T.~Kohonen.
\newblock Improved versions of learning vector quantization.
\newblock pages 545 --550 vol.1, jun 1990.

\bibitem{Kohonen98}
Teuvo Kohonen and Panu Somervuo.
\newblock Self-organizing maps of symbol strings.
\newblock {\em Neurocomputing}, 21(1-3):19 -- 30, 1998.

\bibitem{02Koons}
RD~Koons and J~Buscaglia.
\newblock {Interpretation of glass composition measurements: the effects of
  match criteria on discrimination capability}.
\newblock {\em {JOURNAL OF FORENSIC SCIENCES}}, {47}({3}):{505--512}, {MAY}
  {2002}.

\bibitem{99Koons}
RD~Koons and JA~Buscaglia.
\newblock {The forensic significance of glass composition and refractive index
  measurements}.
\newblock {\em {JOURNAL OF FORENSIC SCIENCES}}, {44}({3}):{496--503}, {MAY}
  {1999}.

\bibitem{01cod}
F~Korn, BU~Pagel, and C~Faloutsos.
\newblock {On the ``dimensionality curse{''} and the ``self-similarity
  blessing{''}}.
\newblock {\em {IEEE TRANSACTIONS ON KNOWLEDGE AND DATA ENGINEERING}},
  {13}({1}):{96--111}, {JAN-FEB} {2001}.

\bibitem{Li2009}
Y~Li, M~Xie, and T~Goh.
\newblock {A study of project selection and feature weighting for analogy based
  software cost estimation}.
\newblock {\em Journal of Systems and Software}, 82:241--252, 2009.

\bibitem{77Lindley}
DV~LINDLEY.
\newblock {PROBLEM IN FORENSIC-SCIENCE}.
\newblock {\em {BIOMETRIKA}}, {64}({2}):{207--213}, {1977}.

\bibitem{09NAS}
Committee on~Identifying the Needs of the Forensic Sciences
  Community;Committee~on Applied and National Research~Council.
  Theoretical~Statistics.
\newblock {\em Strengthening Forensic Science in the United States: A Path
  Forward}.
\newblock National Academies Press, 500 Fifth Street, N.W., Lockbox 285,
  Washington, DC 20055, 2009.

\bibitem{Seheult78}
Allan Seheult.
\newblock On a problem in forensic science.
\newblock {\em Biometrika}, 65(3):646--648, December 1978.

\bibitem{Veenman2005}
CJ~Veenman and MJT Reinders.
\newblock {The nearest subclass classifier: A compromise between the nearest
  mean and nearest neighbor classifier}.
\newblock {\em {IEEE TRANSACTIONS ON PATTERN ANALYSIS AND MACHINE
  INTELLIGENCE}}, {27}({9}):{1417--1429}, {SEP} {2005}.

\bibitem{Walsh94}
K.~Walsh, J.~Buckleton, and C.~Triggs.
\newblock A practical example of the interpretation of glass evidence.
\newblock {\em Science and Justice}, 36(4):213--218, 1996.

\bibitem{09aZadora}
G.~Zadora.
\newblock {Evaluation of evidence value of glass fragments by likelihood ratio
  and Bayesian Network approaches}.
\newblock {\em {ANALYTICA CHIMICA ACTA}}, {642}({1-2, Sp. Iss. SI}):{279--290},
  {MAY 29} {2009}.

\bibitem{09Zadora}
G.~Zadora and T.~Neocleous.
\newblock {Likelihood ratio model for classification of forensic evidence}.
\newblock {\em {ANALYTICA CHIMICA ACTA}}, {642}({1-2, Sp. Iss. SI}):{266--278},
  {MAY 29} {2009}.

\end{thebibliography}

\section{Prototype Learning for Nearest Neighbor Classifiers}

Research in prototype learning is an active field of study \cite{Devi2002, Bezdek2001, Chang74, Dasarathy94, Li2009, Bezdek98, Cano2005, Garain2008, Veenman2005, García2008, Kohonen98, Kohonen90}. A review of the literature in this field has revealed two categories of PLS: 1) instance selection, and 2) instance abstraction. Instance selection involves selecting a subset of instances from the original training set as prototypes. Using what Dasarathy terms as \emph{edit rules}, instance selection can take place in four(4) different ways.

\be
\item incremental (CNN \cite{Hart68a})
\item decremental (RNN)
\item a combination of 1 and 2
\item border points, non-border points or central points
\ee

Instance abstraction involves creating prototypes by merging the instances in a training set according to pre-determined rules. For example, Chang \cite{Change74} merges two instances if the have the same class, are closer to each other than any other instances and the result of merging does not degrade the performance of NNC. The following section gives a brief survey of PLS for both instance selection and instance abstraction.

\subsection{Instance Selection}

\subsubsection{Condensed Nearest Neighbor (CNN)}

\subsubsection{Reduced Nearest Neighbor (RNN)}

\subsubsection{Minimal Consistent Set (MCS)}
In 1994, Dasarthy \cite{Dasarathy94} presented the MCS algorithm. This is a PLS designed to select exemplar samples of a training set with the aim of reducing the computational demands of NNC, while maintaining high consistency levels that is all the original samples are correctly classified by prototypes under the NN rule. Dasrathy's selection process is based on the concept of NUNS, the Nearest Unlike Neighbor Subset. With this concept, he first defines the initial subset as the entire training set, then for each instance in the subset the distance of the NUN is found and all instances with the same class whose distance is closer than the distance of the NUN are stored and counted as votes. The instance with the most votes is then designated as a prototype. All instances which contributed to the number of votes for this prototype is then removed from the other lists. The process is continued until the subset is no longer being reduced (**i need to clear this up**) 


\subsection{Instance Abstraction}
\subsubsection{Chang}
Chang’s work deals with finding prototypes for nearest neighbor classifiers. The basic idea of his algorithm is to start with every sample in the training set as a prototype, then merge any two nearest prototype (p1 and p2 to form p*) with the same class as long as the recognition rate is not degraded. The new prototype p* can be formed by simply finding the average of p1 and p2 or the average vector of weighted p1 and p2. p* will have the same class as the individual prototypes p1 and p2. The merging process will continue until the number of incorrect classifications of patterns in the data set starts to increase. \fig{chang}, shows Chang’s algorithm.

\begin{figure}[h!]
\small
\begin{center}
\begin{tabular}{ p{8cm} }
\hline
\begin{verbatim}
Step 1: Start with an arbitrary point tj
        in B* and assign it to A*
Step 2: For all points tk in B* such that
        class (tk) is not equal to class (tj),
        update bk to be the distance between
        tk ans tj if this daistance is smaller
        than the present bk. Otherwise, bk is
        unchanged.
Step 3: Amoung all the points in B*, find the
        point ts which has the smallest bs
        associated with it.
Step 4: If tj is not the nearest point ts such
        that the classes of tj and ts are
        different, go to Step 6. Othewise
        continue.
Step 5: Check whether or not d(tj, ts) is less
        than bj. If no, go to Step 6. If yes,
        let bj=d(tj, ts) and continue.
Step 6: Let j=s, move ts from B* to A*, and go
        to Step 2 until B* is empty. When B* is
        empty, the final b1,...,bm are the
        desired ones.
\end{verbatim}
 \\ \hline
    \end{tabular}
\end{center}
\caption{Chang's algorithm for finding prototypes}\label{fig:chang}
\end{figure} 
